Author's response to reviews

**Title:** Effects of mental health self-efficacy on outcomes of a mobile phone and web intervention for mild-to-moderate depression, anxiety and stress: Secondary analysis of a randomised controlled trial

**Authors:**

Janine Clarke (janine.clarke@unsw.edu.au)
Judith Proudfoot (j.proudfoot@unsw.edu.au)
Mary-Rose Birch (mpbirch@hammond.com.au)
Alexis Whitton (alexis.e.whitton@gmail.com)
Gordon Parker (g.parker@blackdog.org.au)
Vijaya Manicavasagar (v.maricavasagar@blackdog.org.au)
Virginia Harrison (Gini.harrison@gamil.com)
Helen Christensen (h.christensen@blackdog.org.au)
Dusan Hadzi-Pavlovic (d.hadzi-pavlovic@blackdog.org.au)

**Version:** 4  **Date:** 10 September 2014

**Author's response to reviews:** see over
Mr Carlo Rye Chua  
Journal Editorial Office  
Biomed Central

MS: 4629134881245716 - “Effects of mental health self-efficacy on outcomes of a mobile phone and web intervention for mild-to-moderate depression, anxiety and stress: Secondary analysis of a randomised controlled trial”

Dear Mr Chua

We apologise for the lack of clarity and have now clearly specified the institutional addresses for the authors on the title page as follows:

1Black Dog Institute, Hospital Road, Randwick, NSW 2013, Australia

2School of Psychiatry, UNSW Australia, High Street, Kensington, NSW 2052, Australia

We hope that this editorial revision meets with your approval, and that the manuscript is now acceptable for publication in BMC Psychiatry.

Yours sincerely

Janine Clarke PhD (on behalf of the authors)